

The role of learning in innovation: in-house versus externally contracted R&D experience

Pilar Beneito, M. Engracia Rochina and Amparo Sanchis

Ivie

Los documentos de trabajo del Ivie ofrecen un avance de los resultados de las investigaciones económicas en curso, con objeto de generar un proceso de discusión previo a su remisión a las revistas científicas. Al publicar este documento de trabajo, el Ivie no asume responsabilidad sobre su contenido.

Ivie working papers offer in advance the results of economic research under way in order to encourage a discussion process before sending them to scientific journals for their final publication. Ivie's decision to publish this working paper does not imply any responsibility for its content.

La Serie EC, coordinada por Matilde Mas, está orientada a la aplicación de distintos instrumentos de análisis al estudio de problemas económicos concretos.

Coordinated by Matilde Mas, the EC Series mainly includes applications of different analytical tools to the study of specific economic problems.

Todos los documentos de trabajo están disponibles de forma gratuita en la web del Ivie http://www.ivie.es, así como las instrucciones para los autores que desean publicar en nuestras series.

Working papers can be downloaded free of charge from the Ivie website http://www.ivie.es, as well as the instructions for authors who are interested in publishing in our series.

Edita / Published by: Instituto Valenciano de Investigaciones Económicas, S.A.

Depósito Legal / Legal Deposit no.: V-4044-2009

Impreso en España (octubre 2009) / Printed in Spain (October 2009)

WP-EC 2009-11

The role of learning in innovation: in-house *versus* externally contracted R&D experience

Pilar Beneito, M. Engracia Rochina and Amparo Sanchis^{*}

Abstract

This paper analyses the role of learning in firms' innovation success, taking into account the heterogeneous nature of innovation activities, and in particular, distinguishing between learning arising from the internal organization of R&D activities and learning from externally contracting these activities. We use a representative sample of Spanish manufacturing firms for the period 1990-2006, and within an innovation production function approach, we estimate count data models to investigate the influence of firms' in-house and externally contracted R&D experience in the achievement of innovative results. Our results show that learning is important in the achievement of product innovations when the firms organize R&D activities internally, and that experience from externally contracted R&D activities does not influence the number of product innovations.

Keywords: innovation, accumulation of knowledge, in-house R&D experience, externally contracted R&D experience, count data models. **JEL Classification**: O30, O34, C23, C10.

Resumen

En este trabajo se analiza el papel del aprendizaje en el éxito innovador de las empresas, tomando en consideración la naturaleza heterogénea de las actividades innovadoras, y en particular, distinguiendo entre el aprendizaje que proviene de la realización interna de actividades de I+D y el aprendizaje que proviene de la contratación externa de estas actividades. Para este trabajo se utiliza una muestra representativa de empresas manufactureras en España durante el período 1990-2006, y dentro del marco de una función de producción de innovaciones, se estiman modelos "count" con el fin de investigar la influencia que tiene en la obtención de resultados innovadores la experiencia que proviene de la I+D realizada dentro de la empresa y de la contratada externamente. Nuestros resultados muestran que el aprendizaje tiene un papel importante en la obtención de innovaciones de producto cuando las empresas organizan sus actividades de I+D internamente, y que la experiencia que proviene de la contratación externa de actividades de I+D no influye sobre el número de innovaciones de producto.

Palabras clave: innovación, acumulación de conocimiento, experiencia en I+D interna, experiencia en I+D contratada externamente, modelos para datos "count".

^{*} P. Beneito: University of Valencia. M.E. Rochina and A. Sanchis: University of Valencia and ERI-CES. Corresponding author: erochina@uv.es

1. INTRODUCTION

Despite widespread acknowledgement of the importance of continuity in the performance of R&D activities for the achievement of innovation outcomes, the empirical literature has devoted little attention to the role of learning, understood as the accumulation of knowledge, in the process of innovation.

In this paper we follow Beneito *et al.* (2007) and argue that R&D learning, understood as the accumulation of knowledge and measured as past experience in the performance of R&D activities, is an important driver in the achievement of innovation results, and that its effect is not properly measured by R&D capital stock.¹ However, in this paper we extend their analysis by considering that knowledge accumulation and learning derived from the performance of in-house R&D activities, is of a different nature as compared to externally contracted R&D.² In particular, our working hypothesis is that in-house R&D experience relevant in the achievement of product innovations than the firm' R&D experience related to externally contracted R&D activities. According to Mowery (1983), the performance of R&D activities in-house is usually related to complex research projects requiring knowledge of a highly specialized, idiosyncratic variety, specific to a firm, or knowledge involving a high degree of coordination within the

¹ The usual approach in the literature to capture the concept of knowledge capital and its cumulativeness nature has been the "knowledge capital" model of Grilliches (1979). This model considers that, similar to physical capital, and using the "perpetual inventory method", knowledge capital is accumulated from period to period at a linear and constant rate proportional to R&D investments, subject to a constant depreciation rate.

² The complexity of the process of innovation and the heterogeneous nature of R&D activities has been extensively analysed in the literature. Within the approach of the evolutionary theory of technological innovation, the multiplicity of R&D activities performed by firms has been described by the concepts of technological trajectories (Pavitt, 1984) or technological regimes (Nelson and Winter, 1982, Malerba and Orsenigo, 1990).

firm. On the other hand, extramural performance of R&D activities entails research projects requiring more generic knowledge, applicable to a relatively wide range of industries and firms, and dealing with isolated or separable aspects of a firm's operations. Consequently, one may expect product innovations to be more related to the learning process and accumulation of knowledge associated with the in-house performance of R&D activities, as compared to the extramural performance of these activities.

The aim of this paper is to investigate whether firms' R&D effectiveness, i.e., the rate at which R&D investments yield innovation output, depends both upon firms' accumulated in-house and externally contracted R&D experience, respectively. In particular, we test the hypothesis that in-house R&D experience, measured as the number of years devoted to the performance of in-house R&D activities, is more important for the achievement of firms' product innovations than externally contracted R&D experience.

We use for this purpose a representative sample of the population of Spanish manufacturing firms for the period 1990 to 2006. The dataset is drawn from the *Encuesta sobre Estrategias Empresariales* (ESEE, henceforth), a survey carried out annually since 1990 providing detailed information at the firm level. We estimate, within the framework of an innovation production function and using count data models, the influence of firms' accumulated in-house and externally contracted R&D experience on their R&D innovative effectiveness, measured as the number of product innovations. In order to do this, and following Beneito *et al.* (2007), we treat R&D experience as a moderator variable and, distinguishing between in-house and externally contracted R&D.

The main contribution of this paper to the existing empirical literature is that, to the best of our knowledge, this paper is the first attempt to empirically address the role

of in-house and externally contracted R&D experience respectively, in the achievement of product innovations. To anticipate our results we obtain that, after controlling for R&D capital stock and other firms' individual heterogeneity factors, in-house R&D experience matters for the achievement of product innovations, i.e., that the number of years devoted to the performance of in-house R&D activities affects positively to the number of product innovations introduced by firms. We also find that externally contracted R&D experience plays no role in the achievement of product innovations. In addition, we also obtain that the performance of informal innovation-related activities, and the discontinuity in the performance of R&D activities, are significant determinants of the number of product innovations introduced by firms.

The rest of the paper is organised as follows. Section 2 briefly describes our theoretical framework and related literature. Section 3 presents the data. Section 4 discusses the empirical model and section 5 presents the estimation results. Finally, section 6 concludes.

2. THEORETICAL FRAMEWORK AND RELATED LITERATURE

The importance of knowledge accumulation in explaining innovation has been developed by the approach of evolutionary theory (Nelson and Winter, 1982). The argument is based on the idea that experience allows the accumulation of knowledge, which is associated with dynamic increasing returns in the form of learning-by-doing and learning-to-learn effects. This stream of literature considers that innovations are the result of a process of accumulation of firms' specific competencies (Rosenberg, 1976). In particular, by investing in R&D projects, firms develop abilities in the form of knowledge, both scientific and informal know-how, that may be used to develop further innovations at consecutive times. According to this view, firms benefit from *dynamic increasing* *returns* in the form of learning-by-doing, learning-to-learn or scope economies in the production of innovations (Cohen and Levinthal, 1989).³

The accumulation of knowledge firms obtain from experience in performing R&D activities is likely to affect positively the achievement of innovation outcomes, as stressed by Nelson (1982, page 462):

"Strong knowledge means ability to guide R&D effectively. Stronger knowledge enables a larger expected advance to be achieved from a given R&D outlay: alternatively, strong knowledge reduces the expected cost of any R&D achievement. Strong knowledge enhances efficiency both by enabling R&D to proceed on a generally better set of candidate projects, and by enabling the set worked upon to reflect more accurately particular demands and needs."

And regarding the sources of knowledge:

"Knowledge is not only won through specialized knowledge-seeking activities; knowledge is also won as by-product of searching for new technologies. Knowledge of correlates and of effective testing techniques grows through experience. One learns about efficacious R&D strategies through one's successes and failures. What succeeded and fails last time gives clues as to what to try next, etc. The applied R&D system itself generates new knowledge as well as new techniques." (Nelson, 1982, p. 463).

In order to understand the process of knowledge accumulation, it is useful to characterize R&D activities as iterated cycles of problem-solving, in which organizations select a problem, device a set of potential solutions, and test and choose the optimal option (Newell and Simon, 1972). These cycles of problem-solving build up experience in relevant fields and raise the firms' stock of knowledge (Nelson, 1982, Dosi and Marengo, 1993). As firms accumulate experience and relevant knowledge, the effectiveness of their

³ For a review of this literature, see Dosi and Marengo (2007) and references therein.

research and selection processes improves. Experience in previous research projects turns out to be important in at least three categories of knowledge (West and Iansiti, 2003): which problems are more important to solve, improved understanding of the search process and tools, and information about the most likely potential solutions. These sources of knowledge can be considered as different forms of firms' learning.⁴

The theory of "absorptive capacity" by Cohen and Levinthal (1989) may also be used for the foundation of the role of experience in R&D activities. They suggest that R&D "not only generates new information, but also enhances the firm's ability to assimilate and exploit existing information" (Cohen and Levinthal, 1989: 21). ⁵ By investing in R&D, and therefore by accumulating R&D experience, firms develop their ability to identify, assimilate and exploit externally available knowledge, that is, what they call a firm', "learning" or "absorptive" capacity. This absorptive capacity represents a sort of learning that differs from learning-by-doing: while learning-by-doing refers to the mechanism by which firms become more efficient as they accumulate experience in doing what they are already doing, absorptive capacity allow firms to assimilate outside knowledge in doing new different things. Therefore, the accumulation of knowledge firms obtain from experience in performing R&D activities allow firms to develop their

⁴ The literature of organizational learning also emphasizes the key role of experience in improving organizational performance. According to this literature, the production process creates knowledge about the organization of production that enhances the firm future productivity. The accumulation of this knowledge, or learning, is associated mainly with new technologies or new plants, and gives rise to what is called as organizational capital. This organizational capital, or experience, is usually measured as accumulated output (see, e.g. Bahk and Gort, 1993, and Jovanovic and Nyarko, 1995, and references therein). For instance, Bahk and Gort (1993) introduce experience into the production function as a factor that influences productivity, and construct two measures of the stock of experience: cumulative output since firm birth, and age of the firm.

⁵ See Cohen and Levinthal (1990) for a discussion of the cognitive structures underlying learning.

absorptive capacity and thus is likely to affect positively the achievement of innovation outcomes.

Regarding related empirical literature, there is a wide body of empirical literature that has focused on the analysis of the relationship between firms' R&D input (measured as R&D capital stock, R&D expenditures, or as the ratio of R&D expenditures to sales or revenues) and innovative output (measured, e.g., in terms of patents or productivity). The relationship between innovation, R&D and patents has been surveyed by Griliches (1990), who reports a robust R&D-patents relationship at the firm level. Among the most well known works are those of Schmookler (1966, ch. 2), Scherer (1965), Bound et al. (1984), Hausman et al. (1984), Hall et al. (1986), Pakes and Griliches (1984), Scherer (1983) and Acs and Audretsch (1989). See also Henderson and Cockburn (1993), Branstetter (1996) and Crépon et al. (1998). Another strand of the literature has been devoted to the analysis of innovation persistence per se, both in the achievement of innovations (see, e.g., Geroski et al., 1997, Malerba et al., 1997, Cefis, 2003) and in the performance of R&D activities (Máñez et al., 2009, Peters, 2007). These empirical studies, however, have not directly modelled R&D experience as an additional driver of innovation success.

More recently, a number of empirical works have analysed, using CIS surveys, the innovative performance of firms by relating innovation inputs to innovation outputs. Some of these works are Klomp and van Leeuwen (2001) for the Netherlands, Smith and Sandven (2001) for Norway, Lööf and Heshmati (2001) for Sweden, or Mairesse and Mohnen (2005) and Kremp and Mairesse (2004) for France. However, these empirical studies do not explicitly take into account the possibility that the effectiveness of the R&D innovation inputs changes as firms accumulate experience in the performance of their R&D activities. In particular, there is a lack of empirical evidence explicitly analysing the role of firms' experience in R&D activities as a key driver of their

innovative success. To the best of our knowledge, only West and Iansiti (2003) and Beneito et al. (2007) consider the role of experience in R&D as a key driver of R&D performance. In the context of the evolutionary theory of organizational competencies, West and Iansiti (2003) provide evidence for the US semiconductor industry that experience accumulation and experimentation are two organizational tools that generate flows of new knowledge that through the learning process significantly affect firms' performance. However, their empirical analysis is limited to a reduced number of research projects in one particular industry, and their measure of experience is rather simple and limited: they use a dummy variable indicating if at least one of the project members involved in technology selection decisions has experience in the research organization. On the other hand, Beneito et al. (2007), using a representative sample of Spanish manufacturing firms for the period 1990-2002, provide evidence that the performance of R&D activities, that is, R&D experience, matters in the achievement of innovation results. However, in their analysis of how R&D experience matters for the effectiveness of R&D capital, they do not take into account the heterogeneous nature of R&D activities, and in particular, they do not distinguish between in-house and externally contracted R&D experience as different sources of learning, and so as two different ways of performing R&D that may have potentially different roles in the achievement of innovations, and this is the aim of our paper.

3. THE DATA: R&D EXPERIENCE, R&D STRATEGY, AND PRODUCT INNOVATIONS.

The data used in this paper are drawn from the ESEE for the period 1990-2006. This is an annual survey that is representative of Spanish manufacturing firms classified by industrial sectors and size categories.⁶ It provides exhaustive information at the firm level

⁶ The sampling procedure of the ESEE is the following. Firms with less than 10 employees were excluded from the survey. Firms with 10 to 200 employees were randomly sampled, holding around 5% of the

on a number of production and market issues, including information on innovation activities.⁷ Regarding product innovations, the particular question in the ESEE is as follows: "Indicate if during year t the firm obtained product innovations (either completely new products or with so important modifications that they are perceived as different from the previous ones). If yes, indicate its number".

In this section, we present some descriptive statistics that are calculated conditioning to firms that declare to perform R&D at least one year in the sample, and that report information both on the product innovation question and on all the variables involved in estimation. Applying these criteria we end up with a sample of 12,598 observations, corresponding to an unbalanced panel of 1853 firms.

Table 1 lists and describes the variables involved in estimation, according to the following criteria.⁸ Regarding the inputs in the innovative process and as formal sources of innovation, the ESEE provides information not only on firms R&D expenditures, but also on whether R&D activities are internally organized within the firm or are externally contracted. The ESEE also reports information on firms' informal innovation-related activities, which may also affect positively to the achievement of innovation results. These informal activities include, following Beneito (2003, 2006): services of scientific and technical information, works oriented to normalization and quality control, efforts to assimilate imported technologies, marketing studies, design, and other activities.⁹

population in 1990. All firms with more than 200 employees were requested to participate, obtaining a participation rate equal to around 70% in 1990. Important efforts have been made to minimise attrition and to annually incorporate new firms with the same sampling criteria as in the base year, so that the sample of firms remains representative of the Spanish manufacturing sector over time.

⁷ See http://www.funep.es/esee/ing/i_esee.asp for further details.

⁸ Table 1 also includes a number of variables that are used as control variables in estimation: age and firm size, industry and year dummies.

⁹ The information in the ESEE about these informal activities is collected on a 4-years basis.

Table 1. Variables definition

Product innovations	Number of product innovations introduced by the firm during the year.
R: R&D-capital	The knowledge capital derived from the firm's R&D investment follows the historical or perpetual inventory method:
	$R_{it} = (1 - \delta) R_{it-1} + I_{it-1}$
	where δ is the rate of depreciation, <i>R</i> is the R&D-capital stock and <i>I</i> are real
	R&D expenditures (current R&D has been deflated using industrial prices
	for the whole manufacturing industry).
	To calculate the R&D-capital according to the equation above we need an
	initial value for 1 to start the recursion. We use for that purpose the
	activities. By backwards induction, the sequence of past R&D expenditures
	can be imputed till the first year of R&D activities, when the initial R&D-
	capital stock is equal to zero. The R&D-capital is defined for a depreciation
	rate of 15 percent and a pre-sample growth rate of real R&D investment
	equal to the mean growth rate for the firms which perform R&D activities and are observed during the sample period, that is $a = 4.5\%$
F. R&D experience	Number of years the firm has been investing in R&D in the past.
Lized personnel in "t"	Dummy variable taking value 1 if the firm has recruited (during current
Hiteu personner in t	vear) personnel with experience in corporate R&D. Information on this
with R&D experience	variable is only available from 1998 onwards.
Previous number of	Previous to t number of periods where the firm has not performed R&D (it
spells of R&D inactivity	accounts for the number of intermittencies in which the firms has not
	performed R&D in the past).
Duration of the last spell	Length in years of the last (before η spen of R&D macuvity.
of R&D inactivity	
Scientific/technical	scientific and technical information and 0 otherwise
services	
Quality control	Dummy variable taking value 1 if the firm has undertaken works of
Imported technology	Dummy variable taking value 1 if the firm has undertaken efforts to
imported technology	assimilate imported technologies, and 0 otherwise.
Marketing	Dummy variable taking value 1 if the firm has undertaken marketing
_	studies orientated to the commercialisation of new products, and 0
Design	Dummy variable taking value 1 if the firm has undertaken design activities
Design	and 0 otherwise.
Other	Dummy variable taking value 1 if the firm has undertaken other informal
	innovation-related activities, and 0 otherwise.
Age	Age of the firm.
Age squared	Age of the firm to the square.
Size1	Dummy variable that equals 1 if the number of employees of the firm is
0:0	above 10 and below or equal to 20, and 0 if otherwise.
Size2	above 20 and below or equal to 50, and 0 if otherwise.
Size3	Dummy variable that equals 1 if the number of employees of the firm is
	above 50 and below or equal to 100, and 0 if otherwise.
Size4	Dummy variable that equals 1 if the number of employees of the firm is
0. 5	above 100 and below or equal to 200, and 0 if otherwise.
Sizes	above 200 and below or equal to 500, and 0 if otherwise.
Size6	Dummy variable that equals 1 if the number of employees of the firm is
5220	above 500, and 0 if otherwise.
Industry dummies	Industry dummies accounting for 20 industrial sectors of the NACE-93
	Classification.
Time dummies	The dummes accounting for the 17 years in our sample period.
Exclusive externally	strategy uniquely on externally contracted R&D
contracted K&D strategy	states, anquer on externary contracted here.

With respect to our measure of innovation output, firms obtain product innovations in 32.52% of the sample observations (therefore, we need to take into account the presence of a high number of zero counts in product innovations for the econometric analysis in section 4).10 Of these observations, an 81.60% correspond to firms that declare to perform R&D activities and, within this percentage, a 92.49% are observations corresponding to firms that carry out in-house R&D activities (either combined or not with externally contracted R&D activities), and the remaining 7.51% of the observations correspond to firms performing only externally contracted R&D activities. Regarding the 67.48% sample observations where firms do not introduce any product innovation, a 47.59% of these observations correspond to firms that perform R&D activities. Of these, 82.97% of the observations correspond to firms performing inhouse R&D activities (again either combined or not with contracted R&D activities), and the remaining 17.03% correspond to firms that perform only externally contracted R&D activities. Regarding informal innovation-related activities, for observations in which firms introduce product innovations, in 86.99% of the cases they perform at least one of the informal innovation-related activities, whereas this percentage is 70.51% for observations in which firms do not introduce product innovations.¹¹

¹⁰ Regarding firms, instead of firms' observations, in our sample 67.38% of firms introduce at least one product innovation along the sample period.

¹¹ More in detail, for observations in which firms introduce product innovations, in 43.54% of the cases there are involved services of scientific and technical information, in 66.93% works oriented to normalization and quality control, in 30.73% efforts to assimilate imported technologies, in 42.79% marketing studies, in 58.53% design, and in 3.39% other activities. For observations in which firms do not introduce product innovations these percentages are as follows: in 26.29% of the cases are involved services of scientific and technical information, in 52.65% works oriented to normalization and quality control, in 20.72% efforts to assimilate imported technologies, in 33.88% design, and in 2.11% other activities.

We now turn into the analysis of the patterns of continuity in the performance of R&D activities. Our central hypothesis is that the accumulation of knowledge and learning firms obtain from experience in performing R&D activities is likely to affect positively to the achievement of innovation results. Thus, one may expect the effect of experience on innovation results to depend upon the degree of continuity in the performance of these activities, that is, upon the degree of continuous accumulation of knowledge over time. To explore this issue, we show in Table 2 the distribution of frequencies for the number of firms in our sample according to the following criteria: total number of accumulated years of R&D experience (irrespective of being continuously accumulated or with periods of R&D inactivity in between), total number of spells of R&D inactivity spells. As can be seen in Table 2, 90% of firms in our sample have less than 12 years of total accumulated R&D experience, less than 2 spells of R&D inactivity, and less than 5 years as average duration of the R&D inactivity spells.

In order to describe the link between experience and continuity in R&D activities, and innovation results, measured as the number of product innovations, we perform, with a merely descriptive purpose and following a sequential approach, negative binomial regressions of the number of product innovations on the following variables:¹² total number of accumulated years of firm R&D experience, total number of firm R&D inactivity spells, and average duration of the R&D inactivity spells. Our aim is exploring the potential correlation between the aforementioned variables and the number of product innovations the firm introduces yearly, rather than searching for a causal effect. The results of this analysis are reported in Table 3. As it is shown, the total number of accumulated years of firm R&D experience has a positive and significant correlation with the yearly number of product innovations the firm introduces (Regression 1). However, a

¹² The negative binomial method of estimation will be explained in detail is section 4.

Table 2. Experience and continuity in Rob activities.						
Percentage of firms	Percentage of firms	Percentage of firms				
according to the total	according to the total	according to the				
number of	number of R&D	average duration of				
accumulated years of	inactivity spells	R&D inactivity spells				
R&D experience						
	55.29	55.29				
31.00	33.75	13.40				
13.78	9.27	8.52				
10.33	1.50	6.20				
7.14	0.19	3.63				
5.82		3.01				
5.26		2.69				
5.57		1.50				
3.13		1.50				
3.13		0.75				
2.38		0.69				
2.00		0.63				
1.44		0.50				
2.19		0.50				
0.69		0.38				
1.13		0.75				
1.57		0.06				
3.44						
	Percentage of firms according to the total number of accumulated years of R&D experience 31.00 13.78 10.33 7.14 5.82 5.26 5.57 3.13 3.13 2.38 2.00 1.44 2.19 0.69 1.13 1.57 3.44	Percentage of firms according to the total number of accumulated years of R&D experiencePercentage of firms according to the total number of R&D inactivity spells31.0033.7531.0033.7513.789.2710.331.507.140.195.825.265.573.133.132.382.001.442.190.691.131.573.444				

Table 2. Experience and continuity in R&D activities.

Table 3. Descriptive negative binomial regressions for productinnovations

Regression 1	Coefficient	P-value
Total number of accumulated	0.086	0.000
years of firm R&D experience		
Regression 2		
Total number of firm R&D	-0.141	0.468
inactivity spells		
Regression 3		
Average duration of all the	-0.127	0.000
firm R&D inactivity spells		
Regression 4		
Total number of accumulated	0.057	0.014
years of firm R&D experience		
Total number of firm R&D	-0.040	0.808
inactivity spells		
Average duration of firm R&D	-0.086	0.000
inactivity spells		

Table 4. R&D experience by R&D strategy, and product innovation results

		A
Intervals of R&D	Yearly average number of product	Yearly average number of product
experience (years)	innovations for firms with mainly	innovations for firms with mainly an
	an in-house R&D strategy	externally contracted R&D strategy
1 – 3 years	4.06	1.95
4 – 6 years	5.17	2.73
7 – 9 years	6.54	5.62
10 – 12 years	9.16	10.43
13 – 17 years	4.01	2.62
Total	4.60	2.35

negative and significant correlation is found with the average duration of firms' R&D inactivity spells (Regression 2), and a negative but not significant correlation with the total number of firms' R&D inactivity spells (Regression 3). In Regression 4 we include the three variables simultaneously and find qualitatively the same results than in the sequential regressions. However, the positive correlation for the total number of accumulated years of R&D experience, although still significant, has been reduced from 0.086 to 0.057. Performing a Hausman test (Hausman, 1978) between the results from Regression 1 and Regression 4 we obtain a chi-squared statistic with a value of 104.36 and a p-value of approximately 0. Therefore, we clearly reject the hypothesis of the two coefficients being equal in the two regressions.

The above results suggest that in order to explain the role of accumulated R&D experience on the number of product innovations, in estimation we need to control simultaneously for variables characterising the patterns of discontinuity in R&D activities, otherwise, our results about accumulated R&D experience could be biased. Among the variables we have considered, the one that turns out to be more relevant is the duration of the firm R&D inactivity spells. In section 3, and for estimation purposes, as control variables for each period t we will include both the number of previous R&D inactivity spells, and the duration of the previous R&D inactivity spells. To avoid endogeneity work this variable has been proved to be a better proxy for depreciation of knowledge than the *average* duration of the previous R&D inactivity spells). To avoid endogeneity problems and, therefore, to take properly into account causal relationships, we include in estimation the previous to t values for these variables.

Once we have already noticed that even controlling for discontinuity in the performance of R&D experience, accumulated years of R&D experience still seem to matter, we want to distinguish between two types of R&D experience (associated with inhouse or externally contracted R&D activities, respectively). For this purpose, we

establish a typology of firms according to the type of R&D activities they mainly perform. We construct two dummy variables taking value one when either the majority of years the firm performs R&D activities it performs in-house R&D activities (and zero otherwise), or when the firm performs externally contracted R&D activities (and zero otherwise), respectively. One firm is defined as having mainly an in-house R&D strategy either when the firm only performs such type of R&D activities, or when the majority of years the firm is classified as performing in-house R&D activities (see the definition in more detail in section 4). In the opposite case, the firm is classified as a firm having mainly an externally contracted R&D strategy. According to this classification, approximately 73% of firms in our sample are following mainly an in-house R&D strategy obtain a yearly average of approximately 5 product innovations, whereas firms with an externally contracted R&D strategy obtain a yearly average of 2 product innovations.

Finally, using jointly the information about the accumulative nature over time of knowledge through R&D experience and our typology of firms' R&D strategy, we report in Table 4 the averages of the number of product innovations that firms achieve each year when they are in their 1^{st} to 3^{rd} year of R&D experience, in the 4^{th} to 6^{th} year of R&D experience, and so on. Regarding firms performing mainly an in-house R&D strategy, 90% of them accumulate less than 13 years of R&D experience. As for firms performing mainly an externally contracted R&D strategy, 90% of them accumulate less than 9 years of R&D experience. Looking at these 90% of the distributions, we see in Table 4 that the average number of product innovations that firms yearly achieve rise with R&D experience. For the group of firms with an in-house R&D strategy (externally contracted R&D strategy) this average number ranges from 4.06 (1.95) in the first three years of R&D experience to 9.16 (5.62) in the interval of 10 - 12 years of R&D experience (7 – 9 years), where it

is reached the 90% of the distribution. Therefore, for the 90% of the distributions, the average number of product innovations is larger for firms with an in-house R&D strategy than for firms with an externally contracted one. The externally contracted R&D strategy seems to be in general a less productive strategy in terms of the number of product innovations firms obtain.

4. EMPIRICAL MODEL AND ECONOMETRIC PROCEDURE

The focus of this paper is to measure the extent to which in-house and externally contracted R&D experience matters in determining the effectiveness of R&D activities, that is, in the achievement of innovation results. Our approach is based on the concept of an *innovation production function* that may be expressed in general as follows

$$N_{it} = f(x_{it}, \beta)$$
(1)

where *i* refers to the firm and *t* to the time period, N_{it} stands for any chosen indicator of innovation results, and x_{it} represents the vector of innovation inputs in the equation. Usual components of x_{it} are R&D inputs, quite often measured by R&D capital. Following Beneito *et al.* (2007), our innovation production function will differ from the standard one in that the effectiveness of R&D capital is specified as a function of the R&D experience of the firm. In particular, the parameter vector β may be decomposed as

$$\beta = [\beta_1(\mathbf{E}_{it}), \beta_2]$$
⁽²⁾

where β_1 is the parameter measuring the "innovative effectiveness" of the R&D input, E_{it} stands for firms' R&D experience, and β_2 stands for other inputs' parameters. Therefore, the effect of R&D in the achievement of innovation outcomes depends on R&D experience, measured as the number of years the firm has been engaged in R&D activities. In particular, for our empirical estimation we assume that expression (1) takes the form

$$N_{it} = A(t) R_{it}^{\beta_1(E_{it})} \exp(z_{it}\beta_2)$$
(3)

where R_{it} is knowledge or R&D capital (derived from the flow of real R&D investments),¹³ E_{it} is the firm's R&D experience, and z_{it} stands for an index of other inputs and control variables. Expression (3) includes a direct proportionate relationship between the R&D capital and innovation counts moderated by a multiplicative set of variables hypothesized to shift the distribution of expected innovation results.

The econometric approach to estimate equation (3) is conditioned by the kind of data used to measure innovation success, that is, the output of the innovation process (N_{it}) . In this paper, innovation output will refer to the number of product innovations introduced by the firm during the period under analysis. This means that our dependent variable is of *count data* type: event counts (non-negative integers) for unit *i* during time period *t*, and in any given year many firms may not introduce innovations, so that we may have a high number of zero counts in our sample.

It is standard in the literature to assume that the Poisson distribution is a reasonable description for such count data. According to the Poisson process, research results are the outcome of an unknown number of Bernoulli trials with a small probability of success. The basic Poisson probability specification is

$$\Pr(N_{it} = n_{it}) = f(n_{it}) = \frac{e^{-\lambda_{it}} \lambda_{it}^{n_{it}}}{n_{it}!}$$
(4)

We may model the single parameter of the Poisson distribution function, λ , as a function of our explanatory variables, *x*, and parameters, β , in the standard fashion

$$\lambda_{\rm it} = \exp(\mathbf{x}_{\rm it}\beta) \tag{5}$$

It is easily shown that

$$E[N_{it}|x_{it}] = Var[N_{it}|x_{it}] = \lambda_{it} = exp(x_{it}\beta)$$
(6)

¹³ For a discussion on the use and construction of the R&D capital, see, for example, Hall and Mairesse (1995). Details about how we construct this measure are given in Table 1.

so that λ_{it} represents the arrival rate of innovations per firm per year, and also the expected number of innovation outcomes per firm per year. Taking logs in (6) we get

$$\log E[N_{it}|x_{it}] = \log \lambda_{it} = x_{it}\beta$$
(7)

If the explanatory variables are used in logs, the estimated β are the elasticities of the expected number of innovations with respect to these variables. In our case, we assume that expression (6) takes the form

$$\lambda_{it} = A(t) R_{it}^{\beta_1(E_{it})} \exp(z_{it}\beta_2)$$
(8)

One restrictive assumption of the Poisson model is the assumption that the variance of N_{it} equals its mean (see (6)). As Cameron and Trivedi (1998) note, the Poisson regression fails if there is unobserved heterogeneity in the data. Neglecting unobserved heterogeneity leads to overdispersion and excess of zeros. In the presence of such overdispersion, though the estimated parameters may still be consistent, their standard errors will typically be under-estimated, leading to spuriously high levels of significance. The heterogeneity can be modelled as follows:

$$\exp(\mathbf{x}_{it}\beta + \mu_i) = \left[\exp(\mathbf{x}_{it}\beta)\right]\eta_i \tag{9}$$

where η_i is a random term with $E[\eta_i]=1$. The Negative Binomial model (NB) can be derived as a Poisson mixture that assumes a gamma distribution for η_i . In the NB case $E[N_{it}]=\lambda_{it}$ and $Var[N_{it}|\mathbf{x}_{it}]=\lambda_{it}+\phi\lambda^{2-\kappa}$, where κ is assumed to be 1 or 0 in most empirical applications.¹⁴ Therefore, the NB model nests the Poisson model, and it is possible to test one specification against the other by testing the significance of the parameter ϕ in estimation.

¹⁴ If *k*=0 then we have $Var[N_{it}|\mathbf{x}_{it}] = \lambda_{it} + \phi \lambda^2$, which is the default case estimated by the Stata programme that we use for estimation.

Although the NB model allows for overdispersion, it has been noted by Gurmu (1997) that it provides poor fit if there are *excess zeros* in the data, that is to say, when the incidence of zero counts is greater than expected by the Poisson and standard NB distributions. This may occur if the zeros and positive observations are assumed to be generated by the same process when this is not the case. An alternative approach to deal with the likely different nature of the zeros and the positive values is the *zero inflated model*. The zero inflated model gives more weight to the probability that the count variable equals zero. It considers an underlying mechanism to distinguish between what could be named "non-innovators" and "potential innovators" ("*non-users*" and "*potential users*" in other fields of the empirical literature), with probability $q(w_a\gamma)$ and $1-q(w_a\gamma)$, respectively, where w_a represents the vector of variables to be used for estimating these probabilities. The zero inflated model can be estimated for the Poisson and the NB distribution (ZIP and ZINB models, respectively), and, in estimation, we can use the *Vuong statistic* (Vuong, 1989) to test the non-nested ZIP or ZINB models against their Poisson or NB counterparts.¹⁵

The probability function for the zero inflated model is defined as:

$$P_{it}^{ZIP}(n_{it}/x_{it}) = 1(n_{it} = 0)q_{it} + (1 - q_{it})P_{it}^{P}(n_{it}/x_{it})$$
(10)

where $P_{it}^{p}(n_{it}/x_{it})$ stands for the standard Poisson model (the standard NB model in the case of the ZINB model). The zero inflated model jointly estimates two equations: one of them is a binomial probit or logit model to estimate the probability (q_{it}) of a zero against a positive value for the count variable, and the other equation estimates the probability of the observed count according to (10).

¹⁵ The null hypothesis in the Vuong test is that the two models being considered are equally close to the true specification. Rejection of the null hypothesis leads to the acceptance of the zero inflated version of the model.

We have considered that our count data model may be subject to a problem of excess zeros because the mechanism explaining which firms are potential (product) innovators may be different of that explaining the positive number of product innovations. Although in estimation we have selected those firms performing R&D activities at least one year of the sample period, it may be the case that in a given year either the firm is not performing R&D, or its innovation efforts are not aimed at introducing product innovations. In such a case, we will observe a zero count because this firm is not a potential product innovator, which differs from those zero counts of firms that search for product innovations but have not been successful in a given year.

The goal of this paper is to distinguish between the in-house R&D activity carried out by firms, and the externally contracted R&D activities in order to determine if, as expected, in-house R&D activities play relatively a more important role in the achievement of product innovations than externally contracted R&D, given the nature of the accumulation of knowledge associated with each R&D strategy. The basic idea is whether or not the in-house R&D experience has a higher or lower premium than contracted R&D experience in terms of higher chances of introducing product innovations.

As our baseline Model, we start estimating equation (3) for the case where R&D experience in a given year *t* is measured as the sum of the number of past years the firm has been performing R&D activities, without specifying if these R&D activities are carried out in-house or are externally contracted (we refer to this case as Model I in our table of estimation results). As stated above, the specification given by equation (3) means that the impact of R&D capital on the rate of product innovations is assumed to be a function of the R&D experience of the firm. This function may be non-linear, so in order to allow for a non-linear relationship we assume the following quadratic form:

$$\beta_1(E_{it}) = \alpha_0 + \alpha_1 E_{it} + \alpha_2 E_{it}^2$$
(11)

Formally, β_1 is defined as the percentage change in innovation output generated by a one percent change in R&D capital. Thus, this elasticity represents the effectiveness of R&D capital, moderated by R&D experience, in obtaining innovation outputs. Note that α_0 would be the standard elasticity parameter if R&D experience would not matter for R&D success. In addition, α_1 captures the impact of firm's R&D experience on R&D effectiveness, and α_2 is the change in the impact of firm's R&D experience on R&D effectiveness. If the estimate of α_2 is significantly positive (negative), then the relationship between R&D effectiveness and firm's R&D experience approximates to a "U-type" (inverted "U-type"). However, if the estimate of α_1 is significantly different from zero but the estimate of α_2 is not significant, then firm' R&D effectiveness is a monotonically increasing or decreasing function of firm's R&D experience.

In order to distinguish between in-house and externally contracted R&D experience, as potential and differentiated sources of accumulation of knowledge and learning, and so as affecting differently to the R&D capital effectiveness, we consider in estimation three approaches. These three approaches may be regarded as three different methods of capturing and distinguishing between in-house and externally contracted R&D experience. In the first approach, we make the hypothesis that the in-house realisation of R&D activities is a condition *sine qua non* to accumulate knowledge and learning, whereas externally contracted R&D activities in isolation do not necessarily create such learning effects. To test this hypothesis, we have split the total number of years of R&D experience into two measures: on the one hand, we consider the number of years of in-house performance of R&D activities; on the other hand, we have summed up the number of years the firm only contracts R&D activities, but does not carry out in-house these activities. We call IE_{it} and EE_{it} to our measures of in-house

R&D experience and externally contracted R&D experience, respectively, and specify then the R&D capital elasticity as:

$$\beta_1(E_{it}) = \alpha_0 + \alpha_1 E_{it} + \alpha_2 E_{it}^2 = \alpha_0 + \alpha_1^I I E_{t_i} + \alpha_2^I I E_{it}^2 + \alpha_1^E E E_{it} + \alpha_2^E E E_{it}^2$$
(12)

We refer to this case as Model II in our table of estimation results.

In a second approach, we consider that the contribution of in-house and externally contracted R&D activities to total R&D experience depends on the relative effort devoted to each of these alternatives, measured as the percentage of total R&D expenditure accounted for by each of them. As an example, if a firm allocates in a given year fifty percent of its total R&D investment to in-house R&D activity and fifty percent to contracted R&D, we could say that, in that year, the total R&D experience of that firm is fifty percent in-house R&D experience, and fifty percent contracted R&D experience. Total R&D experience of a firm i in a given year t is computed as a weighted sum as follows:

$$E_{it} = \sum_{\tau=1}^{t} \left(d_{i\tau}^{I} \cdot \rho_{i\tau}^{I} + d_{i\tau}^{E} \cdot \rho_{i\tau}^{E} \right) = \sum_{\tau=1}^{t} d_{i\tau}^{I} \cdot \rho_{i\tau}^{I} + \sum_{\tau=1}^{t} d_{i\tau}^{E} \cdot \rho_{i\tau}^{E} = IE_{it} + EE_{it}$$
(13)

where d_{ir}^{I} and d_{ir}^{E} are dummy indicators taking value 1 if the firm undertakes in-house and externally contracted R&D activities, respectively, in year τ , and where ρ_{ir}^{I} and ρ_{ir}^{E} are the shares of total R&D expenditures devoted to in-house and externally contracted R&D activities, respectively. To test statistically different effects, we allow the coefficients α_{1} and α_{2} to vary for in-house and contracted R&D experience, and then (11) takes the form:

$$\beta_{1}(E_{it}) = \alpha_{0} + \alpha_{1}E_{it} + \alpha_{2}E_{it}^{2} =$$

$$= \alpha_{0} + \alpha_{1}\left(IE_{it} + EE_{it}\right) + \alpha_{2}\left(IE_{it} + EE_{it}\right)^{2} =$$

$$= \alpha_{0} + \alpha_{1}^{I}IE_{it} + \alpha_{1}^{E}EE_{it} + \alpha_{2}^{I}IE_{it}^{2} + \alpha_{2}^{E}EE_{it}^{2} + \alpha^{I\&E}IE_{it} \cdot EE_{it}$$
(14)

We refer to this case as Model III in our table of estimation results.

Finally, we follow a third approach for measuring in-house and externally contracted R&D experience. In this case, we classify firms into two groups according to what we consider to be "mainly an in-house R&D strategy" or "mainly an externally contracted R&D strategy". For this purpose we summed up, on the one hand, the number of years in which a firm performs mainly in-house R&D activities, considering as such those years with only in-house R&D spending, and also those years with a higher percentage of in-house R&D spending as compared to contracted R&D spending. On the other hand, we summed up the number of years in which a firm performs mainly contracted R&D, considering as such those years with only contracted R&D, and those years with a higher percentage of contracted R&D spending as compared to in-house R&D spending. According to these criteria, a firm is classified into the first group (firms with "mainly an in-house R&D strategy") if the number of years doing mainly in-house R&D activities is greater than the number of years doing mainly contracted R&D activities. The firm is classified into the second group (firms with "mainly an externally contracted R&D strategy") if the case is the other way around. As a final step, we multiply total R&D experience by a dummy indicator that identifies firms in one or another group. The specification for our R&D-capital elasticity, expression (11), becomes in this case:

$$\beta_1(E_{it}) = \alpha_0 + (\alpha_1^I E_{it} \cdot d_{iI} + \alpha_2^I E_{it}^2 \cdot d_{iI}) + (\alpha_1^E E_{it} \cdot d_{iE} + \alpha_2^E E_{it}^2 \cdot d_{iE})$$
(15)

where d_{ii} and d_{ik} equal one if firm *i* has been classified into the first or the second group, respectively, as defined above. Note that, in this case, estimated coefficients should be interpreted as the effect of total R&D experience for firms which have mainly in-house R&D experience as compared to the effect of total R&D experience for firms which have mainly externally contracted R&D experience. We refer to this case as Model IV in our table of results. Finally, from 1998 onwards, the ESEE includes information about firms' recruitment of R&D workforce. In particular, the questionnaire of the ESEE asks firms to respond "whether or not the firm has recruited (during current year) personnel with experience in corporate R&D". Thus, we construct a dummy variable capturing this information and introduce this variable into the estimation of Model II.¹⁶ The inclusion of this dummy variable leads us to discard more than half of the sample observations since it is available only since 1998, but we find interesting to include it in estimation because is captures the idea we want to test in this paper: that in-house experience, in this case embodied in hired R&D personnel, contributes noticeably to firms innovation success. We refer to this case as Model V in the table of estimation results.

Taking into account the different specifications given to $\beta_1(E_{it})$ in each of the Models presented above, and taking logs in (8), our estimating function takes the form

$$\log \lambda_{it} = \log A(t) + \beta_1(E_{it}) \cdot \log R_{it} + z_{it}\beta_2$$
(16)

where $\beta_1(E_{it})$ has to be replaced by expression (11), (12), (14) or (15), depending on the Model. As an example, for the particular case of our baseline Model, substituting expression (11) into (3) gives

$$N_{it} = A(t) R_{it}^{(\alpha_0 + \alpha_1 E_{it} + \alpha_2 E_{it}^2)} \exp(z_{it} \beta_2)$$
(17)

and, taking logs,

$$\log N_{it} = \log A(t) + (\alpha_0 + \alpha_1 E_{it} + \alpha_2 E_{it}^2) \log R_{it} + z_{it} \beta_2 = \log A(t) + \alpha_0 \log R_{it} + \alpha_1 E_{it} \log R_{it} + \alpha_2 E_{it}^2 \log R_{it} + z_{it} \beta_2$$
(18)

Control variables in z_{it} include informal innovation-related activities carried out by firms, firm size dummies, firm age and its square, industry dummies accounting for 20 industrial sectors of the NACE-93 classification, and time dummies approximating *log*

¹⁶ We include this dummy variable only in Model II, but conclusions hold irrespective of the model we consider.

A(t). The estimation equation also includes in z_{it} the two variables which account both for the previous number of firms' R&D inactivity spells and for the duration of the last one of these spells. The "zero inflate equation" (which aims at estimating the probability of being a "non-innovator", and which is used to weight the probability of zeros in the data as showed in (10)), includes all the variables that enter z_{it} , as well as a variable that accounts for those firms that follow an R&D strategy based completely on externally contracted activities. This variable has been proved to be a good predictor for zero product innovations in exploratory work.

5. ECONOMETRIC RESULTS

The econometric results from estimation of Models I to V are reported in Table 5. We only present in this table the results corresponding to the ZINB model, that is, the zero inflated negative binomial model, but we have tested all the distributional alternatives described in section 4. In all cases, the parameter capturing overdispersion in the data, ϕ , is statistically significant, indicating the rejection of the Poisson against the NB distribution. In addition, the Voung statistic leads to reject the NB model in favour of the ZINB model. These tests are reported at the bottom of Table 5.

The first column in Table 5 reports the results corresponding to our baseline model (Model I), where a measure of total R&D experience is included without distinguishing between in-house and externally contracted R&D activities. The second, third and fourth columns of Table 5 display the results for our Models II, III and IV described above, respectively, and finally, the fifth column shows Model V, corresponding to the case in which Model II also includes the dummy variable of "*hired personnel in t with corporate R&D experience*". The top half of each column displays the estimation results for our innovation production function and the bottom half of each column presents the results for the zero inflate equation.

Innovations						
	Model I	Model II	Model III	Model IV	Model V	
a	0.065***	0.064***	0.063***	0.070***	0.033	
a_0	(0.015)	(0.015)	(0.015)	(0.019)	(0.029)	
α	0.009***	· · ·	· · ·	, , , , , , , , , , , , , , , , , , ,	· · ·	
<i>a</i> ₁	(0.003)					
	-0 0006***					
α_1	(0.000)					
_	()					
α_1^I		0.010^{***}	0.014^{***}	0.009***	0.007**	
		(0.003)	(0.003)	(0.003)	(0.003)	
α_2^I		-0.0007***	-0.0009***	-0.0006***	-0.0005**	
2		(0.000)	(0.000)	(0.000)	(0.000)	
α^E		-0.001	-0.011*	-0.0035	-0.005	
a_1		(0.007)	(0.006)	(0.004)	(0.008)	
F		0.001	0.001	0.0002	0.0002	
$\alpha_2^{\scriptscriptstyle D}$		(0.001)	(0.001)	(0.0002)	(0.001)	
		()	()	()	()	
$\alpha^{I} \cdot \alpha^{E}$			-0.001			
			(0.001)			
Hired personnel in "t"					0.800***	
with R&D experience					(0.217)	
Previous number of	-0.030	-0.012	0.021	-0.033	0.060	
spells of R&D inactivity	(0.149)	(0.148)	(0.141)	(0.152)	(0.188)	
Duration of the last	-0 074**	-0.069**	-0.067*	-0.061*	-0.080**	
spell of R&D inactivity	(0.035)	(0.034)	(0.034)	(0.034)	(0.040)	
	х <i>у</i>	· · ·	× ,	· · · ·	· · ·	
Scient./Tech. Services	0.118	0.121	0.156	0.152	0.095	
	(0.114)	(0.114)	(0.108)	(0.119)	(0.172)	
Quality control	-0.564***	-0.573***	-0.564***	-0.565***	-0.586***	
Ture in a sub- 1 day -1- in a 1 a sub-	(0.137)	(0.137)	(0.133)	(0.137)	(0.166)	
Imported technology	-0.095	-0.097	-0.103	-0.074	(0.029)	
Mortroting	(0.120)	(0.125)	(0.121)	(0.129)	(0.176)	
Marketing	0.213	0.218	(0.123)	(0.125)	(0, 144)	
Design	0.397***	0.391***	0.395***	0.120)	0 434***	
Design	(0, 117)	(0.116)	(0, 115)	(0.120)	(0, 155)	
Other	-0.324	-0.336*	-0 244	-0.399*	-1 020***	
0 0000	(0.198)	(0.202)	(0.202)	(0.213)	(0.339)	
Age	0.002	0.002	0.002	0.001	0.011	
0	(0.005)	(0.005)	(0.005)	(0.005)	(0.009)	
Age squared	-0.000	-0.000	-0.000	0.000	-0.000	
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	
Size2	-0.039	-0.039	-0.064	-0.087	0.028	
	(0.195)	(0.193)	(0.190)	(0.201)	(0.255)	
Size3	-0.032	-0.016	-0.049	-0.109	-0.226	
	(0.261)	(0.259)	(0.255)	(0.256)	(0.334)	
Size4	0.256	0.249	0.244	0.239	0.158	
~ -	(0.254)	(0.253)	(0.246)	(0.264)	(0.325)	
Size5	-0.010	-0.008	0.003	-0.013	0.136	
	(0.229)	(0.227)	(0.223)	(0.239)	(0.312)	
Sizeo	0.096	0.108	0.084	0.065	0.257	
Constant	(0.253)	(0.253)	(0.244)	(0.267)	(U.376)	
Constant	-0.1/3	-0.050	-0.121	-0.204	-U.U83 (0.520)	
	(0.560)	(0.300)	(0.373)	(0.394)	(0.349)	

Table 5	. Estimates of the	Innovation	Production	Function	for Pr	oduct
		Innovat	ions			

Zero inflate equation					
Exclusive externally	0.731***	0.685***	0.670***	0.734***	0.674***
contracted R&D strat.	(0.151)	(0.156)	(0.155)	(0.157)	(0.213)
Previous number of	0 442***	0 439***	0 440***	0 483***	0 463***
spells of R&D inactivity	(0.088)	(0.088)	(0.088)	(0.098)	(0.109)
Duration of the last	0.084***	0.088***	0.088***	0.083***	0.095***
spell of R&D inactivity	(0.024)	(0.024)	(0.024)	(0.025)	(0.027)
Scient./Tech. Services	-0.396***	-0.403***	-0.390***	-0.351***	-0.535*
	(0.126)	(0.128)	(0.126)	(0.136)	(0.302)
Quality control				0.01011	0.0054
	-0.323***	-0.333***	-0.323***	-0.319**	-0.295*
· · · · · ·	(0.114)	(0.116)	(0.113)	(0.127)	(0.179)
Imported technology	-0.177	-0.192	-0.200	-0.200	-0.046
	(0.143)	(0.146)	(0.145)	(0.156)	(0.280)
Marketing	-0.467***	-0.472***	-0.461***	-0.407***	-0.570**
	(0.146)	(0.147)	(0.144)	(0.151)	(0.233)
Design	-0.552***	-0.549***	-0.545***	-0.552***	-0.691***
	(0.118)	(0.118)	(0.117)	(0.128)	(0.192)
Other	-0.531*	-0.538*	-0.495*	-0.461	-0.361
	(0.292)	(0.296)	(0.291)	(0.340)	(0.521)
Age	0.013**	0.013**	0.013**	0.013**	0.010
	(0.006)	(0.006)	(0.006)	(0.006)	(0.010)
Age squared	-0.000**	-0.000**	-0.000**	-0.000**	-0.000
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Size2	-0.215	-0.216	-0.225	-0.280*	-0.487**
	(0.153)	(0.153)	(0.153)	(0.167)	(0.247)
Size3	-0 414	-0.411	-0 433*	-0.555*	-0 714*
5200	(0, 259)	(0.259)	(0.259)	(0.309)	(0.425)
Size4	-0.193	-0.201	-0 199	-0.223	-0.671**
51201	(0.200)	(0.213)	(0.207)	(0.220)	(0.302)
Size5	0.006	0.003	0.001	0.057	0.206
51265	(0.170)	-0.003	(0.177)	(0.100)	(0.290)
Size	(0.179)	(0.180)	(0.177)	(0.192)	(0.207)
Sizeo	0.028	0.037	(0.022)	0.008	-0.108
	(0.236)	(0.238)	(0.234)	(0.257)	(0.368)
Constant	-0.678*	-0.672*	-0.640*	-0.780*	-0.297
	(0.380)	(0.382)	(0.377)	(0.410)	(0.482)
N. Observations.	12598	12598	12598	11229	5561
Log. pseudo-likelihood	-18480.63	-18471.82	-18452.71	-16920.75	-7832.04
Ho: $\phi = 0$	14.43	14.20	14.19	13.09	11.05
(test overdianorsian)	(P=0.000)	(P= 0.000)	(P=0.000)	(P=0.000)	(P=0.000)
(lest overaispersion)	· · · · ·	. ,		. ,	· · · · ·
Vuong test of ZINB vs.	4.338***	4.342***	4.282***	4.258***	3.778***
standard NB	(0.140)	(0.140)	(0.140)	(0.143)	(0.170)

Robust standard errors in parentheses. * significant at 10%; ** significant at 5%; *** significant at 1% All estimations include 17 time dummies and 20 industry dummies.

In all regressions, the innovation function equation includes (the log of) our R&Dcapital variable and its interactions with R&D experience and with squared R&D experience (see (18)). Depending on the way used to differentiate in-house R&D experience from externally contracted R&D experience, that is to say, depending on whether we look at Model II, Model III or Model IV, the log of R&D capital multiplies expression (12), (14) or (15). For the sake of simplicity, in Table 5 we use the notation $\alpha_1^I, \alpha_2^I, \alpha_1^E$ and α_2^E for the whole set of estimations, taking into account that the interaction terms of the log of R&D capital with in-house R&D experience and contracted R&D experience take different forms in each Model.

If we look at the first column in Table 5, a first result is that both the coefficient α_0 corresponding to the log of R&D capital, and the coefficient α_1 corresponding to the interaction of the log of R&D capital with R&D experience, have positive estimated signs, while the estimated sign of the coefficient α_2 corresponding to the interaction with squared R&D experience is negative. This is a result that confirms previous research by the authors (Beneito *et al.*, 2007), and which suggests that the relationship between R&D capital effectiveness (here expressed in elasticity form) and R&D experience is of an inverted U-type. This means that the effectiveness of R&D capital rises with R&D experience but at a decreasing rate. If we take the (statistically significant) results in this first column, the corresponding R&D-elasticity would be of a magnitude of $\beta_1(E_u) = 0.065$

+ $0.009 \cdot E_{it} - 0.0006 \cdot E_{it}^2$. This means that for a value of 4 years undertaking R&D activities (corresponding approximately to the mean of the sample distribution), the value of the R&D-capital elasticity would be of 0.100, which is almost 30% larger than the elasticity of a firm that has been undertaking R&D for only one year. Figure 1 shows this inverted U-type relationship between the elasticity of R&D capital and R&D experience. Our estimated elasticity reaches its maximum value, approximately, for the 8th year of



0,13 0,12 0,11 0,10 R&D Elasticity 80'0 2000 2000 2000 0,06 0,05 0,04 0,03 2 3 4 567 8 9 10 11 12 13 14 15 16 17 1 R&D experience (years) R&D Elasticity vs. Total R&D Experience ----- R&D Elasticity vs. In-house Experience - R&D Elasticity vs. Contracted R&D Experience

Figure 2. R&D Capital Elasticity vs. In-house and Contracted R&D Experience

R&D experience, and decreases for further years of R&D experience. However, not all points depicted in Figure 1 are equally probable in our sample, and, in particular, by about 85% of the sample distribution lies below a maximum value of 8 years of experience.

Columns II to V show the main results of this paper. The first and main conclusion from these estimations is that, as it has been hypothesized, in-house undertaken R&D activities seem to be the real determinant source for the generation of learning effects during the introduction of product innovations. In Models II and IV the two coefficients that go with the interaction of R&D capital with in-house R&D experience and its square, α_1^I and α_2^I , respectively, are statistically significant and of the same estimated signs as in our baseline model, where total R&D experience was considered. However, the coefficients corresponding to the interaction terms with externally contracted R&D experience, α_1^E and α_2^E , do not render statistical significance in most of the cases. According to the definition of variables in Model II, we could say that only the number of years the firm has been performing in-house R&D counts to explain R&D capital effectiveness, whereas the number of years of externally contracted R&D, if not accompanied by in-house R&D activity, does not seem to help firms to make their R&D capital more productive in terms of product innovations. As for Model IV, we could reach a similar conclusion to the extent that the effectiveness of the R&D capital of firms with "mainly an in-house R&D strategy" is affected by their R&D experience, while firms with "mainly an externally contracted R&D strategy" do not seem to get any return from the number of years they have been performing R&D.

In Figures 2 and 4 we represent the relationship between the elasticity of R&D capital and the R&D experience arising from both in-house and externally contracted R&D activities, respectively. To facilitate the comparison, in the figures we also present the line corresponding to our estimated baseline Model I in Table 5. The figures show

how similar are the results of Models II and IV both with respect to each other and also with respect to the baseline model, although Model IV renders a somewhat higher value for the estimate of α_0 . If we were to choose between one or another model, it seems that, according to the log of the pseudo-likelihood, Model IV provides a better fit to the data.

Only in Model III, when R&D experience is defined as a weighted combination of years of in-house and years of externally contracted experience, the negative estimated sign of coefficient α_1^E turns out to be statistically significant at a 10% level. In the particular case of Model III, the in-house and externally contracted components of total R&D are, respectively, the sum of the number of years doing in-house and contracting R&D activities multiplied by the shares of total R&D allocated to each of them in each year. Then, for those firms performing both in-house and externally contracted R&D, an increase in the share of contracted R&D spending would imply a lower share of in-house R&D. Then the estimated negative sign of α_1^E may be indicating that those firms which intensify their strategy of outsourcing R&D obtain a lower R&D capital elasticity than those which do not do so. However, the weak statistical significance of the estimated coefficient casts doubt on the confidence we should put on this result. Figure 3 shows the relationship between the R&D capital elasticity and the components of R&D experience as defined in Model III.

The last column of Table 5 displays the results for Model V, which corresponds to Model II including the dummy variable, available only since 1998, that accounts for firms that *have recruited personnel with experience in corporate R&D during the current year*. This variable has a positive and highly significant effect on the achievement of product innovations, a result that reinforces the two main conclusions of our paper: first, that R&D experience is an important source of knowledge which matters to explain innovation results, and second, that it is the in-house performance of R&D activities what allows to exploit the effects of learning through experience.



Figure 3. R&D Capital Elasticity vs. a weighted measure of R&D Experience

Figure 4. R&D Capital Elasticity for firms with mainly In-house or mainly Contracted R&D Experience



Although the variable accounting for the number of spells of R&D inactivity does not render statistical significance, the negative and significant effect of the variable that accounts for the duration of the last of these spells is noticeable in all our estimation models. To give an example, a firm that accumulates ten years of past R&D experience but stopped previously during three years, obtains a lower number of product innovations than a firm with ten consecutive and uninterrupted years of R&D activity. This result suggests that, not only the continuity in R&D activities is necessary to achieve learning effects through experience, but also that the longer the periods of R&D inactivity, the higher the depreciation of the acquired learning.

Other complementary results in Table 5 are those related to informal innovationrelated activities. In the estimation of the innovation production function offered in the top half of Table 5, we observe that both marketing and design activities contribute positively to the achievement of product innovations, whereas quality control activities exhibit a negative and significant sign. Also at a descriptive level, our sample data indicates that quality control activities are negatively correlated with the number of product innovations obtained.

We now turn to comment briefly the results of our zero inflate equation, reported at the bottom half of Table 5. Recall that what we are estimating here is the probability of observing zeros, so that a positive sign of a parameter estimate means a higher probability of a zero, and a negative sign means a higher probability of observing a positive number of product innovations. A first result is that the dummy variable accounting for those firms which base their R&D strategy uniquely on externally contracted R&D helps significatively to predict the event of no product innovations. Also the previous number of spells of R&D inactivity of the firm, and the duration of the last one of these spells are, as expected, positively correlated with the observation of a zero outcome. As for informal innovation-related activities, almost all of them are negatively

and significantly correlated with the event of a zero product innovation, reinforcing the hypothesis that these activities correlate positively with the innovative performance of firms. The only exception is imported technology, that it is not statistically significant. Finally, age explains, at a decreasing rate, the probability of zeros, indicating that younger firms are more product innovators than older ones. Firm size, in general, is not significant and, when it is, it indicates a lower probability of a zero outcome for firms of size above the "10 to 20 employees" interval, which is the firm size category left out of the estimation.

6. CONCLUSIONS

In this paper we have tested two hypotheses related to firms' innovation activities using a representative sample of Spanish manufacturing firms (ESEE) for the period 1990-2006. The first one is that, due to knowledge cumulativeness, the period of time during which firms performs R&D activities, which we call R&D experience, is a key determinant of the number of innovations they may achieve. In particular, we have tested the hypothesis that the effect of R&D-capital stock in the achievement of product innovations depends on R&D experience. Our second hypothesis, which constitutes the main aim of this paper, has focused on the investigation of whether firms' R&D effectiveness, i.e., the rate at which R&D investments yield product innovations, depends both upon firms' accumulated in-house and externally contracted R&D experience, i.e., the number of years devoted to the performance of in-house R&D activities, is more important for the achievement of firms' innovation success (measured as product innovations), than externally contracted R&D experience. To test hypothesis we have estimated, within the framework of a knowledge production function and using count data models, the influence of firms' accumulated R&D experience on their R&D innovative effectiveness, measured as the number of product innovations, distinguishing between the performance of in-house and externally contracted R&D activities.

The results of our empirical analysis indicate that, after controlling for R&Dcapital stock and other firms' individual heterogeneity factors, the number of product innovations introduced by firms rises with in-house R&D experience, that is, with the accumulation of technical skills and knowledge that emerges as firms invest in in-house R&D over time. However, the experience that firms accumulate in the performance of externally contracted R&D does not seem to affect the number of product innovations introduced by firms. This result is probably due to the nature of research related to extramural R&D activities, usually of a generic character and not specifically related to the development of new products, which usually requires firm' specific and complex knowledge, arising from a dynamic, cumulative process of in-house R&D activities. Finally, and in addition to past R&D experience, the performance of some informal innovation-related activities and the continuity in the performance of R&D activities have also been found to be important determinants in the achievement of product innovations.

Our results contribute to a better understanding of the channels linking firms' investment in different types of R&D activities and innovation results and thus, of the role of the cumulative process of learning in the effectiveness of R&D investments. These findings may suggest the direction of potential policy measures to be implemented in order to stimulate the production of R&D knowledge. In particular, given that in-house R&D experience positively affects the achievement of product innovations, our results indicate the convenience of implementing policy measures aimed at inducing firms to continuously engage in in-house R&D activities, such as technological policy measures planed within a medium run perspective, or measures aimed at creating a stable institutional framework. More generally, by contributing to understand the heterogeneity

across firms regarding the performance of R&D activities and the achievement of innovation results, our findings may be used as a guide for the allocation of R&D subsidies, a major issue in public policy. Finally, our results are also interesting from a strategic management point of view and may also be used to provide managerial recommendations for several key decisions relating the performance of R&D activities, such as the selection of an appropriate schedule for the different types of organising R&D that may be used to achieve innovation results. We have obtained that organising R&D activities in-house positively affects the probability of obtaining product innovations, thus suggesting how better to manage R&D activities to produce innovations. If we consider innovations as closely related to the production of new knowledge, firms that innovate will also be firms concerned with the management of their produced knowledge. This knowledge from innovation may be considered as a firm's asset and therefore should be managed strategically in order to maximize the returns from innovation, as in the case of other assets such as plant, equipment or brand names.

References

Acs, Z.J. & Audretsch, D.B., (1989), 'Patents as a measure of innovative activity', *Kyklos*, 42(2), 171-80.

Arrow, K.J., 1962, 'The economic implications of learning by doing', *The Review of Economic Studies*, 29, 3, 155-173.

Bakh, B. H., & Gort, M., 1993, 'Decomposing learning-by-doing in new plants', *Journal of Political Economy*, 101, 561-583.

Beneito, P., 2003, 'Choosing among alternative technological strategies: an empirical analysis of formal sources of innovation', *Research Policy*, 32, 693-713.

Beneito, P., 2006, 'The innovative performance of in-house and contracted R&D in terms of patents and utility models', *Research Policy*, 35, 502-517.

Beneito, P., Rochina, M.E. and Sanchis, A., 2007, 'R&D experience and innovation success', IVIE WP-EC 2007-10.

Bound, J., Cummins, C., Griliches, Z., Hall, B.H., & Jaffe, A. (1984) 'Who does R&D and who patents?', in Z. Griliches (ed.), *R&D*, *Patents and Productivity*, Cambridge, University of Chigaco Press, 21-54.

Branstetter, L., 1996, 'Are knowledge spillovers international or intranational in scope?. Microeconometric evidence from the U.S. and Japan', NBER Working Paper Series, No. 5800.

Cameron, A.C. & Trivedi, P.K. (1998), *Regression Analysis for Count Data*, Cambridge: Cambridge University Press.

Cefis, E., 2003, 'Is there persistence in innovative activities?', International Journal of Industrial Organization, 21, 489-515.

Cohen, W.M. & Levinthal, D.A., 1989, 'Innovation and learning: the two faces of R&D', *The Economic Journal*, 99, 569-596.

Cohen, W.M. & Levinthal, D.A., 1990, 'Absorptive capacity: a new perspective on learning and innovation', *Administrative Science Quarterly*, 35, 128-152.

Crépon, B., Duguet, E., & Mairesse, J., 1998, 'Research, innovation and productivity: an econometric analysis at the firm level', NBER Working Paper Series 6696.

Dosi, G. & Marengo, L., 1993, 'Some elements of an evolutionary theory of organizational competences', in England, R.W., (ed.), *Evolutionary Concepts in Contemporary Economics*, University of Michigan Press, Ann Arbor.

Dosi, G. & Marengo, L., 2007, 'On the evolutionary and behavioural theories of organizations: a tentative roadmap', *Organization Science*, 18, 491-502.

Geroski, P.A., Van Reenen, J. & Walters, C.F., 1997, 'How persistently do firms innovate?', *Research Policy*, 26, 33-48.

Griliches, Z., 1979, 'Issues in assessing the contribution of R&D to productivity growth, *Bell Journal of Economics*, 10, 92-116.

Griliches, Z., 1990, 'Patent statistics as economic indicators: a survey', *Journal of Economic Literature*, 28, 1661-1707.

Gurmu, S. (1997) 'Semi-parametric estimation of hurdle regression models with an application to medic aid utilizations', *Journal of Applied Econometrics*, 12: 225-242.

Hall, B.H., Griliches, Z., & Hausman, J., 1986, 'Patents and R&D: is there a lag?', International Economic Review 27(2), 265-283.

Hall, B.H. & J. Mairesse, 1995, 'Exploring the relationship between R&D and productivity in French manufacturing firms', *Journal of Econometrics*, 65, 263-293.

Hausman, J.A., 1978, 'Specification tests in econometrics', Econometrica, 46, 1251-1271.

Hausman, J., Hall, B.H. & Z. Griliches, 1984, 'Econometric models for count data with and application to the patents-R&D relationship', *Econometrica*, 52(4), 354-383.

Henderson, R., & Cockburn, I., 1993, 'Scale, scope and spillovers: the determinants of research productivity in the pharmaceutical industry', NBER Working Paper Series 4466.

Jovanovic B., & Nyarko, Y., 1995, 'A Bayesian learning model fitted to a variety of empirical learning curves, *Brookings Papers on Economic Activity*, 247-299.

Klomp, L. & van Leeuwen, G., 2001, 'Linking innovation and firm performance', *International Journal of the Economics of Business*, 8 (3), 343-364.

Kremp, E., & Mairesse, J., 2004, 'Knowledge management, innovation and productivity: a firm level exploration based on French manufacturing CIS3 data', *NEBR Working Paper* 10237, Cambridge, MA.

Levitt, B., & March, I., 1988, 'Organizational learning', Annual Review of Sociology, 14, 319-40.

Lööf, H. & Heshmati, A., 2001, 'Knowledge capital and performance heterogeneity: a firm level innovation study', *International Journal of Production Economics*, 0, 1–25.

Mairesse, J. & Mohnen, P., 2005, 'The importance of R&D for innovation: a reassessment using French survey data', *The Journal of Technology Transfer*, 30(1–2), 183–197.

Malerba, F. and Orsenigo, L., 1999, 'Technological entry, exit and survival: an empirical analysis of patent data', *Research Policy*, 28, pp. 643-660.

Malerba, F., Orsenigo, L. & Peretto, P., 1997, 'Persistence of innovative activities', sectoral patterns of innovation and international technological specialization' *International Journal of Industrial Organization*, 15, 801-826.

Máñez, J.A., Rochina-Barrachina, M, Sanchis-Llopis, A. & Sanchis-Llopis, J.A., 2009, 'The role of sunk cost in the decision to invest in R&D', *Journal of Industrial Economics, forthcoming.*

Mowery, D.C., 1983, 'The relationship between intrafirm and contractual forms of industrial research in American manufacturing, 1900-1940', *Explorations in Economic History*, 20, 351-374.

Nelson, R.R., 1982, 'The role of knowledge in R&D efficiency', Quarterly Journal of Economics, 97, 453-470.

Nelson, R. & Winter, S., 1982, An Evolutionary Theory of Economic Change, Cambridge, Harvard University Press.

Newell, A., & Simon, H.A., 1972, *Human Problem Solving*, Prentice Hall, Englewood Cliffs, NJ.

Pakes, A., & Griliches, Z., 1984, 'Patents and R&D at the firm level: a first look' in in Z. Griliches (eds.), *R&D*, *Patents and Productivity*, Cambridge: University of Chigaco Press, 55-72.

Pavit, K., 1984, 'Sectoral patterns of technological change: towards a taxonomy and a theory', *Research Policy*, 13, 343-373.

Peters, B., 2007, 'Persistence in innovation: stylised facts and panel data evidence', Journal of Technological Transfer, forthcoming.

Rosenberg, N., 1976, *Perspectives on Technology*, Cambridge University Press, Cambridge.

Smith, K., & Sandven, T., 2001, 'Innovation an economic performance at the enterprise level', *Innovation and Enterprise Creation: Statistics and Indicators*, European Commission, Luxembourg, 106-116.

Scherer, F. M., 1965, 'Corporate inventive output, profits and growth', *Journal of Political Economy*, 73 (3), 240-297.

Scherer, F. M., 1983, 'The propensity to patent', International Journal of Industrial Organization, 1, 107-28.

Schmookler, J., 1966, Invention and Economic Growth, Cambridge: Harvard University Press.

Vuong, Q., 1989, 'Likelihood ratio tests for model selection and nonnested hypotheses', *Econometrica*, 57, 307–333.

West, J. & Iansiti, M., 2003, 'Experience, experimentation, and the accumulation of knowledge: the evolution of R&D in the semiconductor industry', *Research Policy*, 32, 809-825.

PUBLISHED ISSUES*

WP-EC 2008-01 "La orientación al mercado como determinante de la internacionalización de las nuevas empresas" A. Blesa, M. Ripollés, D. Monferrer. March 2008. WP-EC 2008-02 "El balance actuarial como indicador de la solvencia del sistema de reparto" M.C. Boado-Peñas, C. Vidal. March 2008. WP-EC 2008-03 "La influencia de la cultura sobre la búsqueda de información. El caso de la vivienda para 'turismo residencial' en la Costa Blanca". F.J. Sarabia, A. Kanther, J. F. Parra. April 2008. WP-EC 2008-04 "Optimal CEO compensation and stock options". A. Jarque. April 2008. WP-EC 2008-05 "Un análisis bayesiano de la variación temporal del escenario de compra de los hogares". C. Berné, P. Gargallo, N. Martínez, M. Salvador. May 2008. WP-EC 2008-06 "Trade, tariffs and total factor productivity: the case of Spanish firms". M. Dovis, J. Milgram-Baleix. May 2008. WP-EC 2008-07 "The negative effects of failed service recoveries". A.Casado, J. Nicolau, F. Mas. May 2008. "Short-term electricity futures prices: evidence on the time-varying risk WP-EC 2008-08 premium". J. Lucia, H. Torró. May 2008. WP-EC 2008-09 "Wage, price and unemployment dynamics in the Spanish transition to EMU membership". K. Juselius, J. Ordóñez. May 2008. "Determinantes de la divulgación de información previsional en España: WP-EC 2008-10 un análisis de las empresas IBEX 35". M.C. Abad, F. Bravo, M. Trombetta. July 2008. WP-EC 2008-11 "Testing capital-skill complementarity across sectors in a panel of Spanish regions" F. Pérez-Sebastián. September 2008. "Edad y tamaño empresarial y ciclo de vida financiero" WP-EC 2008-12 J. Sánchez-Vidal, J.F. Martín-Ugedo. September 2008. WP-EC 2008-13 "The asymmetric effect of endowments on vertical intra-industrial trade' J. Milgram, A.I. Moro. October 2008.

^{*} Please contact Ivie's Publications Department to obtain a list of publications previous to 2008.

- WP-EC 2008-14 "Health shocks, household consumption, and child nutrition" A. Galiano, M. Vera-Hernández. December 2008.
- WP-EC 2009-01 "The effects of immigration on the productive structure of Spanish regions"J. Martín, G. Serrano, F. Requena. February 2009.
- WP-EC 2009-02 "Corporate governance and impression management in annual press releases"B. García, E. Guillamón-Saorín. March 2009.
- WP-EC 2009-03 "Factores determinantes de la salida a Bolsa en España" J.E. Farinós, V. Sanchis. April 2009.
- WP-EC 2009-04 "New challenges in competitiveness: knowledge development and coopetition"
 C. Camisón-Zornoza, M. Boronat-Navarro, A. Villar-López. April 2009.
- WP-EC 2009-05 "The determinants of net interest income in the Mexican banking system: an integrated model"J. Maudos, L. Solís. April 2009.
- WP-EC 2009-06 "Explaining protectionism support: the role of economic factors" N. Melgar, J. Milgram-Baleix, M. Rossi. April 2009.
- WP-EC 2009-07 "Determinants of interest rate exposure of Spanish banking industry" L. Ballester, R. Ferrer, C. González, G.M. Soto. May 2009.
- WP-EC 2009-08 "Entrepreneurial orientation, organizational learning capability and performance in the ceramic tiles industry"J. Alegre, R. Chiva. June 2009.
- WP-EC 2009-09 "El coste y el desequilibrio financiero-actuarial de los sistemas de reparto. El caso del sistema Español"J.E. Devesa Carpio, M. Devesa Carpio. July 2009.
- WP-EC 2009-10 "Imposición y protección social: efectos sobre el nivel de empleo en la OCDE"
 J.R. García Martínez. July 2009.
- WP-EC 2009-11 "The role of learning in innovation: in-house versus externally contracted R&D experience"P. Beneito, M.E. Rochina, A. Sanchis. October 2009.
- WP-EC 2009-12 "Do process innovations boost SMEs productivity growth? J.A. Máñez-Castillejo, M.E. Rochina-Barrachina, A. Sanchis-Llopis, J.A. Sanchis-Llopis. October 2009.
- WP-EC 2009-13 "Esperanza de Vida Libres de Discapacidad por sexo y Comunidad Autónoma: 2004-2006"
 F.J. Goerlich, R. Pinilla. October 2009.



l vie

Guardia Civil, 22 - Esc. 2, 1° 46020 Valencia - Spain Phone: +34 963 190 050 Fax: +34 963 190 055

Website: http://www.ivie.es E-mail: publicaciones@ivie.es